Author's response to reviews

Title: Awareness of Human papillomavirus and factors associated with intention to obtain HPV vaccination among Korean youth: Quasi experimental study

Authors:

Hae Won Kim (haewon@snu.ac.kr)

Version: 6 Date: 20 January 2015

Author's response to reviews:

Editor's comments:
Please include a separate conclusion section

Answer:
According to the comments, the conclusions were made in the text.
Thank you for these comments.